Author's response to reviews

Title: The relation between insulin resistance and lung function: a cross sectional study

Authors:

Gul Sagun (gulsagun@yahoo.com)
Canan Gedik (drcanangedik@gmail.com)
Esra Ekiz (dr.esra@gmail.com)
Engin Karagoz (dr.engindhly@yahoo.com)
Mumtaz Takir (mumtaztakir@yahoo.com)
Aytekin Oguz (aytekinoguz@hotmail.com)

Version: 7  Date: 4 October 2015

Author's response to reviews:

Dear editor

Page numbers were addressed at the STROBE checklist for each item and I moved the sentence to Author Contributions section indicating that all authors read and approved the final manuscript

Kind regards

4.10.2015

Gul Sagun, MD